Author's response to reviews

Title: Thrombocytopenia in the experimental leptospirosis of guinea pig is not related to disseminated intravascular coagulation

Authors:

Hong-liang Yang (henry7206@hotmail.com)
Xu-cheng Jiang (xjiang@shsmu.edu.cn)
Xiang-yan Zhang (heloise@163.com)
Wen-jun Li (xu_max@hotmail.com)
Bao-yu Hu (xkquo@shsmu.edu.cn)
Guo-ping Zhao (gpzhao@sibs.ac.cn)
Xiao-kui Guo (xkquo@shsmu.edu.cn)

3 4 November 2005

Author's response to reviews:

MS: 135584619713443
Thrombocytopenia in the experimental leptospirosis of guinea pig is not related to disseminated intravascular coagulation
Hong-liang Yang, Xu-cheng Jiang, Xiang-yan Zhang, Wen-jun Li, Bao-yu Hu, Guo-ping Zhao and Xiao-kui Guo

Dear editor,
I am very grateful for your and the referees' comments to the manuscript. According to your comments and requests, we have revised the manuscript, particularly the English use throughout the manuscript version 2.

Reviewer's report:
Minor Essential Revisions
Comment 1: Extensive revision is needed to improve grammar and clarity throughout the manuscript.
Response: Yes, The grammar errors have been checked and corrected. The writing style has been modified throughout the manuscript. And we have gone through the manuscript formatting checklist.

With best wishes,
Yours Sincerely
Xiao-Kui GUO
03/10/2005